Author's response to reviews

Title: Clinical Evaluation Study of the German Network of Disorders of Sex Development (DSD)/Intersexuality: study design, description of the study population, and data quality

Authors:

Anke Lux (Anke.Lux@med.ovgu.de)
Siegfried Kropf (Siegfried.Kropf@med.ovgu.de)
Eva Kleinemeier (kleinemeier@paedia.ukl.mu-luebeck.de)
Martina Juergensen (juergensen@paedia.ukl.mu-luebeck.de)
Ute Thyen (thyen@paedia.ukl.mu-luebeck.de)
DSD Network Working Group (not@valid.com)

Version: 4 Date: 15 April 2009

Author's response to reviews:

Dear Melissa Norton,

We would like to submit the 3rd revised version of the study protocol paper: Clinical Evaluation Study of the German Network of Disorders of Sex Development (DSD)/Intersexuality: study design, description of the study population, and data quality.

Major revisions:
1. We cite the former references 47-66, now 39-58 within the text on page 9.

Minor revisions
2. We removed all unnecessary capitals in the title
3. We put the abbreviations in a paragraph format before the competing interests
4. We removed all vertical rules from the tables.

We hope that the revisions are sufficient. If there are further questions do not hesitate to contact us.

Best regards

Ute Thyen